Author's response to reviews

Title: Epidemiology of rotavirus infection among young children with acute diarrhoea in Burkina Faso

Authors:

Isidore J O Bonkoungou (ouindgueta@gmail.com)
Idrissa Sanou (idrissasanou@yahoo.com)
Fabienne Bon (fabon@u-bourgogne.fr)
Benoit Benon (cliniquetisserins@yahoo.fr)
Sheick O Coulibaly (sheickoumar2@yahoo.fr)
Kaisa Haukka (kaisa.haukka@thl.fi)
Alfred S Traoré (astraore@univ-ouaga.bf)
Nicolas Barro (nicolas_barro@univ-ouaga.bf)

Version: 4 Date: 17 December 2010

Author's response to reviews: see over
Dear editor,

We are very pleased to make final change in the manuscript and submit it for publication into your journal. We have just made some corrections on author’s affiliation details in red.

1. Service de pédiatrie, Centre Médical du Secteur 30 de Ouagadougou, Ouagadougou, Burkina Faso

2. Department of Infectious Disease Surveillance and Control, Bacteriology Unit National Institute for Health and Welfare, P.O. Box 30, 00271 Helsinki, Finland

We could not find the current version of our manuscript in the web links at the bottom of the email and we hope it’s possible to do it.

Yours Sincerely,

Bonkoungou Juste Isidore